Author's response to reviews

Title: Comparison of a one-time educational intervention to a teach-to-goal educational intervention for self-management of heart failure: design of a randomized controlled trial

Authors:

Darren A DeWalt (darren_dewalt@med.unc.edu)
Kimberly A Broucksou (kbroucksou@schr.unc.edu)
Victoria Hawk (victoria_hawk@med.unc.edu)
David W Baker (dbaker1@nmff.org)
Dean Schillinger (dschillinger@medsfgh.ucsf.edu)
Bernice Ruo (bruo@nmff.org)
Kirsten Bibbins-Domingo (kdomingo@medsfgh.ucsf.edu)
Mark Holmes (holmes@schsr.unc.edu)
Morris Weinberger (mweinber@email.unc.edu)
Aurelia Macabasco-O'Connell (aoconnel@sonnet.ucla.edu)
Michael Pignone (pignone@med.unc.edu)

Version: 5  Date: 9 June 2009

Author's response to reviews:

Some typographical errors were corrected.
A few minor word edits were done.
Table 5 text box was made smaller and spacing put in between points 1-4 and description of analysis.
Submitting another copy of Figure 1 because the top box continues to get cut off.
If this version continues to have the first box top line cut off please notify me and I will continue to work with you to get it correct so the box is complete.